Author’s response to reviews

Title: Observational study of a new strategy and management policy for measles prevention in medical personnel in a hospital setting

Authors:

Chang-Pan Liu (joeliu5929@hotmail.com.tw)
Hsi-Peng Lu (lu@mail.ntust.edu.tw)
Tainyi Luor (dr.luor@gmail.com)

Version: 4 Date: 26 May 2019

Author’s response to reviews:

No.: INFD-D-18-01774R3

Title: Observational study of a new strategy and management policy for measles prevention in medical personnel in a hospital setting

BMC Infectious Diseases

Dear editor,

Thank you for reviewing the manuscript entitled “Observational study of a new strategy and management policy for measles prevention in medical personnel in a hospital setting”. Your favorable report is greatly appreciated. I have made essential revisions to the manuscript as requested.

Editors Comments:

1. Funding

This study was supported by grants MMH-107-106 from MacKay Memorial Hospital.

The funders did not have a role in any phase of the study conduct.

2. The manuscript has been uploaded as a single, final, clean version. All relevant tables/figures/additional files have also been uploaded as clean versions.
Yours sincerely,

Hsi-Peng Lu, Ph.D.

Corresponding author,

Department of Information Management,

National Taiwan University of Science and Technology, Taipei, Taiwan.